Agata/AMBMin/May09/final
Minutes AGATA Management Board 

28th – 29th May 2009 Daresbury Laboratory. Start 13:30 28th Finish at 13:00 29th.
Present:
D. Bazzacco

A. Gadea

G. Duchêne
E. Farnea
J Nyberg

J. Simpson 


Project Manager and Secretary

Ch. Theisen
P.J. Nolan


ASC representative

1. Apologies for absence


P. Reiter
2. Minutes of the last meeting 9th – 10th February in Orsay Agata/AMBMin/Feb09/02D

The draft minutes Agata/AMBMin/Feb09/02D were accepted as a correct record of the meeting.

Will be distributed as Agata/AMBMin/Feb09/final.
3. Matters arising from the last meeting and actions outstanding from last phone conference

PR has funds available.

He needs a shopping list.
Action: CT, DB and JN to send a list to PR so we can decide what to buy at the next AMB.

Done see item 7.
4. ASC Report 
P.J. Nolan was welcomed to the meeting as the new Chair of the ASC.
Next ASC is 16-17th June 2009 in Stockholm.

Report on location of AGATA for the second phase.

Meeting with Ganil management took place in January. Learned of the shutdown of Ganil for Spiral which overlap the previous plan for AGATA implementation.
GSI stated that they could take it. Meeting took place in March with GSI management PJN, WK, JJ, JS, JG represented AGATA)
Paper prepared by the ASC sent to GSI, distributed to AMB during the meeting.

Letter received recently from GSI, distributed during the meeting.

GSI has requested the PRESPEC presents a package of proposals and present co-ordination with the wider NuSTAR community.
P.J. Nolan recently contacted by Slyvia Lenzi re a meeting between laboratory directors, PAC chairs, PRESPEC, AGATA and Nustar. 
Any comments and questions to Paul Nolan as soon as possible.

Note need at least 2 years between a location decision and starting at a different laboratory.

New MoU and structure, see item 11.

5. Reports from Working Groups

a. Detector module and characterisation

P. Reiter distributed a report by email which is attached.
Status of return of detectors, cryostat and digitizer from Turkey. Needs signature from Rector of Ankara but should proceed soon. The tax problem has been solved. Shipping to GSI, agreement has to be revalidated every 3 years.

Scanning still an issue and we are behind schedule. May scan C001 in Liverpool even if the one segment at the back is not working.
ATC1. Two capsules improved but red capsule is still rather noisy. Full testing is still to do. 
b. Front end processing
Digitizers

Over temperature shutdown has been implemented and tested successfully. Cut-off at 40oC
Implementation of CFD is needed.
Pre-Processing
VC meeting yesterday.

Week 24, test a set of electronics for another triple.

GTS production started for 20 mezzanines. Delivery will be close to the run so not sure if can use 2nd triple Cluster.

Trigger tested and it is possible to have coincidences between capsules (and anything else that can send a trigger).
Second level trigger in a C-programme.
PCIExpress not yet solved. Bo Cederwall got a licence but for Virtex 5.

Cost of license in $25k.

Could use the RAL scheme for overnight compiling? But some political issues.
See proposal in item 7.

Ongoing work on controls. Still problems with operation and progress with more than one laboratory working on the system. 
CSNSM travel money is an issue. No further funds from IN2P3. Travel required for electronics, DAQ, slow control to get the system going. Need to determine how much extra funding is required.
Issue of Orsay bench test. Agreed needed but is in conflict with setting up of electronics in Legnaro.
What is required in Orsay:

Detector (will have)

Digitizer (plus PS etc.)  One need to be shipped, agreed. 

Fibres. (will need to be purchased)

Two ATCA carriers.

GTS mezzanine missing. Maybe solved end June/July when production modules arrive.

Pizza box (they have?)

Optimistically may be able to have a system in mid July.

Action JS to contact Amel to see what level of travel funds are required.

Action JS to email ASC/P.J. Nolan to request assistance.

Ganil has also made enquiries regarding a similar system.
Open problems.

Effort: 

Damiano contract extended for almost two years. 

GTS effort. Funding for Joel Chavas ends in August. Big problem for triggering and interface with ancillaries. No position or solution has yet been found. 
DN/Padova has to pay back items within Padova.

Funding can be obtained from Cologne. See item 7.
PSA
Grid search implemented. Date from experiment under analysis. Currently at 12keV.

300 events per CPU. Aim of 1kHz (core) rate achieved (3 CPU’s).

Recent meeting, not many attendees. 
Currently have to write traces for the next run. Limited by the basis and testing of algorithms.
c. Data Acquisition
Data Flow

· Week 8 source test

· Some tricky behaviours; Algo libraries debug

· Week 12 beam test 

· No major problems; bug in aws shell (memory leak) : solved

· Note : remote “spying” from Orsay peoples

· May beam test : same as week 12

· June/July beam test :

· Data flow for 2 TC + ancillary

· Merge using event # or time-stamp

· Merge of ancillary

Data Flow: Issues and to do

· Data flow optimisation

· Narval nodes reboot

· ADF

· Some inconsistencies

· Rather slow at FEE producer level

· New requests

· In the DA shopping list :

· Rootifyer

· Watcher templates

· Data Flow meeting next week, Orsay : coordination of algo development and integration
Orsay Meeting week 23 will discuss:

1 - Code organisation 
1-1 Development (policy): usage of svn, tag versions, install directories on site, documentation; 
1-2 Definition of the tasks performed by the different actors 
1-3 Definition of virtual interfaces for algorithms 

2 - Data content 
2-1 Definition of the Data Frames (new versions, new types) and the meta data (configuration Files and/or Frames) 
   + automatic save of conf. used in a run 

3 - Watchers 
3-1 Possibilities 
3-2 What is to be developed 
Administration
· Enrico Calore fellowship expired;  made an excellent job


Hope he will come back for a PhD thesis

· Position of some Orsay peoples was not clear (a short summary of the situation could be : managing and doing the job is not the same)

· Ticket system to handle and track problems. Being tested.

· Does not solve the problem of responsibilities (and manpower to some extend), but hope will help … 

· Visit of Yann + Eric at Legnaro on week 26. Very long todo list. (In general very efficient work when Orsay guys travel)

· Still rather long to do list (user account, disk privilege, default setup, GPFS versus  CNFS, fast link between DAQ box and disks ). 

· To do list for next visit

· NAS installation and configuration

· Switch installation

· Visu pc (new power supply, export home, and system Imager

· Configuring backup system 

· Fixing some Zabbix bug

· Handling VPN key revocation

· Talking with LNL people

· Other stuff (Xav ...) 
Services

· Run Control

· Need coherent implementation (e.g. process_stop) on actor side

· “Clean” destroy and “Ultimate reset” (not only RC stuff)

· Using run number

· Log implementation in actors

· Some zabbix features

· Save used configuration (topology, config, algos, …).

· GUI

· Minor problems found with RC connection

· Rates summary implemented

· GRU

· Some improvement

· New version recently installed

· ftp server (OK)

· Copy to Tier 1 : see Johan’s report

· In general, good feedback from peoples.
Hardware and funds

· Fund issue for CSNSM peoples travel. Additional funds requested. Eric and Nicolas asked to provide schedule their travel for 2009 (?)...  Should I (we) interfere with CSNSM management ? 

· NAS disk server + switches (Sweden) at Legnaro.

· Reminder for the full (6 TC) demonstrator

· 9 pizza : ~ 18 k€
· 60 TB disk (Sun Storage) : 40-60 k€
· Adacore license would help according to Xavier : 21 k€
Slow Control

· Legnaro meeting Feb. 19th. 

· Overview; discussion starting slowly … 

· Koln meeting :

· Global Slow Control needed (config, topology, …)

· Chipscope replacement

· Automatic procedures during setup (including traces analysis)

· Self diagnostic and error handling

· Monitoring

· Orsay (Eric Legay) has the responsibility of Global Slow Control (Fred Saillant from Ganil resigned few days before). Vic Pucknell as an expert consulting

· GSC Draft specifications being discussed (Eric et al)

· Slow Control meeting June 18th, 19th, Orsay :

· WSDL, Topology, Save / Restore, …

· Detailed discussions of functionalities, priorities and planning. 

· First GSC not before end of 2009.

· Rather good communication inside team

Documentation

· Wiki filled slowly. At least useful for DAQ WG.

Remarks

· Less pressure on data flow peoples since first tests were rather positive. Need to encourage the group again.
· Slow Control starting slowly. Nice enthusiasm from Eric.

· Need more coordination on data flow + algo (involves 3WG, at least 5 teams), …

· Implementation schedule for full demonstrator ? (new hardware, installation, …).

d. Ancillary detectors and ancillary detector integration

First production AGAVA delivered. Tested and fully working.
Tests for PRISMA and DANTE are ongoing.
Next step; Integration with Narval. 

GSI have requested AGAVE modules to check integration at GSI.
Poland made AGAVE modules for GSI for PRESPEC. GTS mezzanines are required for these. Note also need support effort required.

At the physics workshop requests for several/many ancillary detectors.

AMB with local group at LNL will need to discuss a sensible schedule.

e. Design and Infrastructure

Meeting on Tuesday this week. Subset of the notes of this meeting:
AXIS Status and plans (Cologne tests June)

PROFIBUS problem to be discussed next week with WAGO and solution to be tested.

Three filter boxes have been produced, and cables in coming days.

Cologne own PLC has been set up with the help of Saclay, but has some problems.  The Saclay PLC can be used as backup.  To be checked further.  

Discussion on AXIS tests at Cologne.  AXIS test should also include HV, and due to small delay, expected tests in week 25.  HV filter installation at GSI on Monday 15th/16th followed by AXIS/HV tests in Cologne.

HV Status & Plans 

All PCBs produced. Minor problems solved. One PCB has been returned to manufacturer with fault.

There is a connector problem to ISEG unit (from PCB to ISEG) and new connectors sought from Germany. End of June expected for completion of HV boxes.

2 HV box mechanics still missing. Need to check possibility of manufacture of more (2 + 2 spares) HV mechanics.
The PT100 connector to be made compatible with HV box input.

Three HV boxes could be left at Cologne after the test.
HV Filters

The HV-filters are ready and GSI are waiting for the Saclay visit "on way to Cologne" to install them in the HV boxes.

Autofill status, VCC and GUI
The Autofill progresses slowly as expected, the date of week 25 possibly would not be kept and somewhat lately it will be installed there, presumably before the summer holiday.

“Summer holidays” in LNL involve technicians and ideal time for VCC installation is week 28-29.  Installation should be in parallel again, but the VCC should again be tested in GSI as fully as possible.

Noted that it is important to test autofill system before holidays start to avoid unexpected problems.

The UPS time for just HV on detectors is approximately 90 minutes. 
Unclear of status of watchdog in Profibus crate.

List of alarms and extra connections (e.g. UPS) ongoing.  Phone dialler being tested.
Grounding

Meeting of 19th June postponed due to Italians being unavailable that day. 
Agreed that the meeting will be held on 8th July.

f. Gamma-ray tracking, simulations and data analysis

GRID EVO minutes circulated by Johan. Need two T1 sites for archiving.
JN and Gaetano will prepare a document of the plan for the next two years.

Items that could be purchased.

* Workstations for online data analysis:

  - 2 workstations; ~2 k€ workstation;
 4 k€
* Grid hardware for T1 Bologna: 

  - 4 x 1 TB harddisk, ~1 k€/TB;

 4 k€

  - 1 tapedrive



16 k€

* Grid hardware for T1 Lyon: 

    - 4 x 1 TB harddisk, ~1 k€/TB;
4 k€

  - 1 tapedrive



16 k€
* Digital oscilloscope;


8 k€
Tracking. Orsay forward tracking implemented.


Online will have tracking but will write PSA hits anyway for offline improvements.


Some work this year to install Strasbourg algorithm in Narval.


Date Bases. Unknown status, need to chase up the actors.
6. Installation of the demonstrator at LNL and project planning

Since AGATA week:

ATC1 refurbishment. Will fix outstanding problem after test this weekend.

ATC1 + cryostat with dummy mounting test. New mechanical parts to get to inter detector distance specification. Now much better but still a small problem. Spare flange sent to Daresbury to investigate problem further. Final report to be sent from the Mechanical Team.

ATC1 on frame and cold.

Cabling problems solved.
Autofill progress slow, awaiting crate.

Beam time. Run this week end, Coulex 58Ni, ATC1, 1 element of Dante. Acquire Dante via segment mezzanines (not AGAVA due to missing GTS). Currently under test. Leak on vacuum chamber, but fixed overnight. Beam will be Saturday and Sunday. Possibility of beam tomorrow. Aims to check timings, correlations and position resolution. 
Next beam time starts 29th June. Difficult to delay this to ensure having two ATCs.  

Priority to try and commission ancillaries. Similar aims to test this weekend.
Financial decision by INFN to shut down LINAC at LNL and to start operation again in February 2010.  Only simple experiments, with the tandem alone, will be considered until then such as AGATA alone…
Proposals should be submitted to Enrico (AMB) on suitability of proposal prior to formal submission to the PAC. This information has been sent to AGATA community.

Enrico will request additional commissioning beam time at the next PAC (14th July) for the period September to end February. These will be limited to tandem beams. Typically ask for 20 days, 4 days per months for this 5 month period.
Request for: AGATA plus DANTE (AGAVA), AGATA plus PRISMA (tandem beams, AGAVA), AGATA with a high multiplicity reaction, AGATA plus a standard plunger, AGATA plus PRISMA (ALPI). 

7. Funding status, capital allocation and running costs

Running costs. 
LNL has just the 20k€ from Italy.
Needs funds as specified in paper submitted to ASC.

26.5k€ from Germany. PR still awaiting news if funded.
Turkey money transfer from GSI?

Running costs from turkey discussed. Agreed to proceed to send to GSI. 

Action: JS to reply to Ayse Turkey transfer to GSI and subsequent transfer to LNL.
Action: JS to summarise position for the ASC. Missing funds in LNL will be 45k€ -20k€ -5k€ (from Germany) = 20k€. Where to get these funds from? Turkey, Bulgaria, Romania and Poland?
Capital funding available:

Peter Reiter (Cologne) has 101k€ (including VAT)


Sweden 15k€


Turkey 80k€ (to be transferred to IPNO for pre-processing and DAQ)


Turkey 40k€ (to be transferred to GSI for running costs)


Turkey ~100k€

Items agreed to be purchased:

PCI Express 20 k € (DB/PR)


4k€ workstations (Sweden)


8k€ oscilloscope (Sweden)


15k€ Padova items (DB/PR)


45k€ Disks (Germany)


Action: CT to contact PR with details of disks to buy.

Action: DB to PR re the items in Padova

Action: DB and PR to try and sort out PCIExpress problem.

Action: JN to contact Bo re purchases in Sweden.
Remaining items to not specifically allocated at present:

20k€ GRID Bologna

20k€ GRID Lyon

18k€ Pizza boxes

21k€ Adocare liscence

xk€ for disks
8. Next AGATA week


GSI January plus launch of collaboration council.


Action: PJN discuss collaboration council at the ASC.

AGATA physics meeting, report on data analysis and to discuss proposals for July 2010 PAC in Turkey, before mid May.
Action: AG to discuss with Sean Freeman to agree if sensible and with Ayse if possible.

9. Location of AGATA for second physics campaign


See ASC report.
10. AGATA Organisation, funding and responsibilities

Discussion of paper submitted by PJN to AMB for discussion.

Discussion of EDAQ meeting.

Papers on proposals by EDAQ team not received. 
No progress or agreement on how to proceed yet.

Need for a project definition for the next phase.

Need for a technical co-ordinator?

Define what is required, where are developments needed?
Host laboratory responsible persons needed.
Issues of single suppliers discussed.

Need to get the costs down (detectors, electronics, cryostats etc. etc.)


Composition of the AMB. Introduction of new people, phased introduction?

New project manager?

Realistic commitments for the next phase needs to be specified.


Action: All to send any thoughts, ideas etc. to Paul Nolan ASAP and prior to the ASC meeting.

Future commitment of new capital:


Italy has ~80k€ for digitizers but should this be committed on existing digitizers?


It is possibly to order 6 digitizers (4 UK and 2 Italy) of the existing design now.

This will fix the LNL phase to end 2010 as total of 24 digitizers.

Agreed to proceed with this option.


Action: DB to contact Patrick Coleman-Smith regarding an order for 2 digitizers.

11. Project Manager’s Report
A report is required for the ASC.
Action: JS will draft next week and request input.

Action: all to send talks to JS. 

Action: AG to send JS information on the physics workshop.
12. AOB

Overheads for general presentation of AGATA status. Continue to distribute.


Talks and presentations:
E Farnea gave a talk at Dubrovnik conference.

E Farnea Varena in June


G Duchêne Strasbourg Summer School in July

G Duchêne Ganil workshop in September


W. Korten Nuclear Chemistry Gordon Conference.


Web. No progress. Needs chasing up. CT and JN to discuss and take up if time later in the year.
13. Date and location of next meeting and dates of phone conferences.



3rd - 4th September in Paris area. GD to arrange a room, most likely in Orsay.
Phone conferences / Lyon system (GD to organise)


8th June 14:00 CET

30 June 10:15 CET

10 July 15:00 CET
3 Aug 10:15 CET
Annex 1 

Status of AGATA Detectors, Deliveries and available Detectors 

Customer acceptance tests


· Detector A005 : Delivery 18/11/08 – Accepted 01/02/09,done at Saclay

· Detector A006 : Delivery 28/11/08 – Accepted 01/04/09, no report sent yet, done at Saclay

· Detector B001 : Delivery 22/01/09 – Rejected 06/02/09 due to leakage current, done at Cologne

· Detector B006 : Delivery 05/03/09 – mounted in ATC3 first for CAT, leakage current in ATC3 ,  2nd CAT test in Daresbury test cryostat. Confirmed results of ATC3. Showed leakage current on core, segment F5 out of specs. Resolution depends on up/down position of cryostat. No report sent yet, done at Cologne

· Detector  C006 : Delivery 30/04/09 – energy resolution of core and all segments well within specs. Crosstalk measurement ongoing. Detector is looking promising and will be needed for the missing C-type detector in ATC3. Done at Cologne

Status of AGATA triple detectors

· ATC1 

      1st modification (at Legnaro, March, Bart + CTT) : preamps modified – fft cleaner, 

2nd modification (at Legnaro, end April, Andreas + CTT + Ebert): cold part grounding       changed -  fft clean (only 10Mhz from accelerator).

Present status: Improved performance with respect to high frequency oscillations and disturbances. One segment missing after cooling down. Date for repair to be agreed.

· ATC2: 

· Exchange of broken Ganil preamps for Milano preamps (due to lack of Ganil spare    parts). HF oscillations (52Mhz) reappear with Milano preamps. 20 new Ganil preamps arrived: With Ganil preamps no oscillations are observed. Milano – effect needs to be understood. Common preamp and detector group meeting in June on problems with preamps and micromatch connectors. 

Common tests with infrastructure group from Saclay of AXIS modules scheduled for week 25 (17 june – 19 June). Date for delivery of ATC2 to be agreed.

· ATC3: 

Two broken detector B006 and C003 dismounted both due to leakage current. Assembly is depending on next available B- and C-type detector. Next B-type detector is B001 which is announced to be delivered in spring 2009. Potential C-type detector is C006. 

Meanwhile ATC3 is used for further R&D work: To increase reliability of cabling in cold part new capton cables were designed and manufactured for cold part. (See picture below.)   
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First test with capton cables performed: LN2 consumption and temperature on 1 detector     measured: Feedthough 16( compared to 18( with standard single wire cables. Test is ongoing.

· Symmetric Triple Cryostat: 

Test in preparation with S002 for capacity-voltage measurement (LCR meter) and  leakage current measurement (pAmp meter). 

Scanning

Activities at Liverpool:

· C001 repair by local (Liverpool) people. Several cycles were done, however every time a few missing segments were found after cooling down. Repair ongoing.

Acticities at Orsay:

· A005 or A006 are tested and ready to be mounted into test cryostat for scanning at Orsay. Mounting of A-type detector into test cryostat will be done at Saclay.










